Author's response to reviews

Title: Life-threatening biopsy of an iliopsoas pseudotumour in a patient with haemophilia: a case report

Authors:

Azan S Al Saadi (azanalsaadi@gmail.com)
Ali H Al Wadan (amresam@hotmail.com)
Samir A EL Hamarneh (hamarneh@hotmail.com)
Mohamed E Emad (mohammede77@hotmail.com)

Version: 9 Date: 10 April 2008

Author's response to reviews:

Revision
Abstract was modified into structured abstract and was resubmitted into the following sections:
Introduction
Case presentation
Conclusions
Figures
Figure 2 and 3 were improved